Author's response to reviews

Title: Are parents' knowledge and practice regarding immunization related to pediatrics' immunization compliance? A mixed method study

Authors:

Omer Qutaiba B. Allela Dr (omarallela2010@gmail.com)
Mohd B. Bahari (wakbaidi@yahoo.com)
Harith K Alqazaz (harithqazaz73@yahoo.com)
Muhammad R.M. Salih (muhanad_rmk@yahoo.com)
Ramadan M Elkalmi (edriph@gmail.com)
Shazia Q Jamshed (Pharmacist1992@live.com)

Version: 3
Date: 15 November 2013

Author's response to reviews: see over
Authors’ Responses to Editor

We are very thankful for such constructive comments which have significantly enhanced our manuscript entitle “Are parents’ knowledge and practice regarding immunization related to pediatrics’ immunization compliance? A mixed method study”

Editor comments:
I would like to see still a few corrections/comments added to the manuscript

Author response: We are very thankful for you and agree to respond to the editors’ comments.

Author’s action: Changes made as suggested.

Sincerely,

The Corresponding Author
Dr. Omer Q.B. Al-Lela
Assistant of Professor in Clinical Pharmacy
School of Pharmacy
Faculty of Medical sciences
University of Duhok (UOD)
Duhok / Iraq
<table>
<thead>
<tr>
<th>#</th>
<th>Comments of editor</th>
<th>Response</th>
</tr>
</thead>
<tbody>
<tr>
<td>Editor</td>
<td>Page 10 - 3rd paragraph - I don’t think inactive immunization is the same as passive immunization. Please, leave just the latter (twice in the paragraph)</td>
<td>Revised and corrected</td>
</tr>
<tr>
<td>Editor</td>
<td>I believe there might have been some translation problem with question number 10, discussed in the last paragraph of p. 11. One thing is to be aware of the potential for (harmful) side-effects of vaccine. Another is to consider vaccination harmful. I can’t agree that this assertion can be considered correct. Could you correct/comment this 'nuance'?</td>
<td>Revised and corrected</td>
</tr>
<tr>
<td>Editor</td>
<td>Page 13 - first paragraph, penultimate line: again, it seems that media is being used for internet, which I think is the correct word to be included. Media would include all other means of communications just described on the previous lines</td>
<td>Revised and corrected</td>
</tr>
<tr>
<td>Editor</td>
<td>I would assume that parents in general are willing to vaccine their children, if we observe the answers to questions 1, 11, 12 and 13. As pointed out by one of the reviewers, although I understand that it has not been particularly studied here, there must be in the discussion some indication that there are barriers to vaccination in Irak that should also be addressed. Maybe your other study on the subject can be referred.&quot;</td>
<td>Revised and my studies added as reference</td>
</tr>
</tbody>
</table>